\nocite{Berlioux.sep.79.295}
\nocite{Berlioux.sep.79.301}
\nocite{Berlioux.sep.80.617}
\nocite{Berlioux.sep.80.349}
\nocite{Ecker.sep.80.333}
\nocite{Berlioux.sep.82.107}
\nocite{Karrenbach.sep.82.265}


\title{Report bibliography}
This is my bibliography of articles
in the progress reports of the Stanford Exploration Project
written from 1974 to the present.
There are  6 
articles.
{\footnotesize
\putbib[SEP]
}
